## PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (http://bmjopen.bmj.com/site/about/resources/checklist.pdf) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below.

## **ARTICLE DETAILS**

TITLE (PROVISIONAL)	Diagnostic Testing and Vaccination for COVID-19 among First Nations, Metis and Inuit in Manitoba, Canada: Protocol for a Nations-Based Cohort Study using Linked Administrative Data
AUTHORS	Enns, Jennifer; Nickel, Nathan C.; Clark, Wayne; Phillips-Beck, Wanda; Sanguins, Julianne; Lavoie, Josée; Romanescu, Razvan; Katz, Alan; Mahar, Alyson; Brownell, Marni; Lix, Lisa; Urquia, Marcelo; Tso Deh, Miyosha; O'Conaill, Carrie; Durksen, Anita; Brownell, Emily

This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non-commercial or government funder and ethics approval.